

Tax haven, pollution haven or both? Thierry Madiès, Ornella Tarola, Emmanuelle Taugourdeau

▶ To cite this version:

Thierry Madiès, Ornella Tarola, Emmanuelle Taugourdeau. Tax haven, pollution haven or both?. International Tax and Public Finance, 2022, 10.1007/s10797-022-09745-x. hal-03772706

HAL Id: hal-03772706 https://hal.science/hal-03772706

Submitted on 24 Oct 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Tax haven, pollution haven or both ?

Thierry Madiès ^{*}, Ornella Tarola [†] & Emmanuelle Taugourdeau [‡]

Abstract

This paper studies the interplay between a poor and a rich country when they compete sequentially over corporate taxes and environmental regulations to attract imperfectly mobile firms. Generally, the poor country undercuts the rich country in terms of corporate taxes and chooses to be both a tax and pollution haven when it is less concerned about the environment than the rich country is and capital integration is low. However, it rarely does better in terms of welfare than the rich country. Finally we find that tax competition immunizes countries against the detrimental effect of globalization on environmental standards.

Keywords: Tax Competition, Capital Integration, Cross-border Pollution, Environmental Standards

JEL classification: H2, R3, R5, Q5.

^{*}Université de Fribourg, Bd de Pérolles 90, CH-1700 Fribourg, Switzerland. Email: thierry.madies@unifr.ch

[†]DISSE, University of Rome La Sapienza, Piazzale A. Moro 5, Rome, Italy. Email: ornella.tarola@uniroma1.it

[‡]Corresponding author: CNRS, CREST, ENS Paris-Saclay, 5 Av Henry Le Chatelier, 91120 Palaiseau, France. Email:emmanuelle.taugourdeau@ensae.fr

1 Introduction

Globalization enhances competition between countries and regions to attract firms, secure jobs, and over time, sustain economic development. Competition between countries and states is multifaceted and public authorities can act through fiscal instruments and regulations such as environmental standards. Policy-makers in high-income countries and in intermediate or low-income countries have a strategic incentive to cut their business taxes and relax environmental standards to attract footloose firms at the expense of their competitors. However, the benefits of economic development have increasingly been brought into question over the last decades, especially in developed countries, on the grounds that a concentration of industrial activities pollutes, not only in the same jurisdiction as the polluting activities, but also in neighboring ones and even in more distant countries, since pollutants can be transported over hundreds, even thousands of kilometers. More generally, global warming, air pollution and acid rain, which are extremely damaging to the environment, are characterized by cross-border spillovers.¹ The purpose of our paper is to present a simple model of competition between a rich (high-income) country and a poor (intermediate or low-income) country. The two countries compete over corporate tax rates and environmental standards in the context of increasing capital mobility and concerns over globalization and cross-border pollution. Policy-makers differ in their environmental awareness. The political trade-off between business- and environmentally-friendly policies depends on a country's level of development. The pressure on environmental issues is higher in high-income countries than in developing or transition countries. For instance, the Standard Eurobarometer survey (2019) clearly shows that public concern about climate change and environmental protection is much lower in Eastern European countries than in EU15 member states. More generally, there is some evidence of an inverted U-shaped income-pollution relationship, the "environmental Kuznets curve" (Chen et al. (2019) for a short survey).

¹There is ample evidence of cross-border pollution between US states (see Millimet (2013) for a comprehensive survey). Transboundary pollution is also well documented in Asia: Japan and South Korea regularly complain that the acid rain they suffer is caused by emissions of sulfur and nitrogen oxides from coal-burning plants in northern China (Abu Sayed, "Cross-border pollution : A growing international problem. The Daily Star, February 19, 2011.). Transboundary air pollution also occurs between European Union member states despite the emission reduction measures adopted under the Convention on Long-Range Transboundary Air Pollution (CLRTAP) and EU legislation (European Environment Agency, 2020).

More specifically, our paper aims to address the following questions: Are poor countries more likely than rich countries to be tax and pollution havens? Are rich countries able to attract capital investment without cutting taxes or relaxing environmental standards? How does international capital mobility affect tax policies and environmental standards? How do differences in environmental awareness between countries affect tax and environmental policies?

To answer these questions, we set up a sequential non-cooperative game in which a rich country and a poor country first set their environmental standards and then compete over corporate taxes to attract internationally (imperfectly) mobile firms. In producing, the firms pollute domestically and generate cross-border pollution spillovers: environmental standards affect pollution and environmental quality both locally and in the competing country. Countries have some leeway when choosing their environmental standards but they have to comply with international environmental agreements. Non-cooperative environmental standard setting does not mean that countries do not cooperate at all on environmental issues. On the contrary, countries often invest considerable efforts into finding cooperative agreements on environmental issues involving transboundary pollution (e.g. the Convention on Long-Range Transboundary Air Pollution and the Kyoto Protocol). However, signatories have a certain level of discretion and strategic room for maneuver in defining their environmental policies (see the above mentioned empirical evidence).

The structure of the game follows from the widely accepted rule that decisions taken in the first stage are irreversible (Pieretti and Zanaj, 2011). This is the case for environmental regulations, which are more binding in the middle-to-long term than tax decisions, which are subject to frequent, sometimes yearly changes. As a result, when setting their environmental standards, countries anticipate that they will also be engaging in corporate tax competition in the second stage of the game. Firms are supposed to bear the mobility costs of relocating from one country to another. In our model, countries maximize tax revenues by imposing a corporate tax that is inversely related to the tax-elasticity of mobile firms. It is straightforward to show that the corporate tax base is less elastic in rich countries than in poor countries. Besides, the lower mobility costs that come with greater economic integration drive corporate taxes down, and richer countries tend to have more integrated economies than poorer ones do. Furthermore, social attitudes towards the environment and pollution differ between countries. A key feature of the model is that by relaxing its environmental standards in the first stage of the game to attract firms, a country may trigger a backlash from the competing country in the form of a corporate tax cut in the second stage. In this model therefore, relaxing environmental standards may not necessarily be the best strategy for the competing countries. Of course, policy-makers can decide to relax their environmental standards and/or cut corporate taxes to attract firms and increase tax revenue. However, this may come at the expense of increased domestic pollution. Indeed, the trade-off depends not only on the relative (political) weights of tax revenues and domestic pollution, but also on pollution spillovers between countries and the extent to which policy-makers are concerned by the latter.

Our main results are the following. The equilibrium environmental standards depend both on domestic concerns about pollution and on concerns for cross-border pollution in the competing country (through a strategic effect). The higher the concern for both types of pollution is, the more stringent the environmental standards are. Interestingly, international capital mobility does not affect the equilibrium environmental standards because governments fully internalize the effects of the latter on firms' location when choosing their corporate taxes in the second stage of the game. Our paper also shows that the poor country generally undercuts the large country and acts as a tax haven because it is more sensitive to corporate inflows. At equilibrium however, the poor country almost always does worse than the rich one. The balance ultimately depends on the level of capital integration and the respective degrees of environmental awareness. Under some circumstances moreover, the rich country may have incentives to undercut the poor one. We also find that higher capital mobility narrows the tax gap between the rich and the poor country. The effect on the respective payoffs ultimately depends on the level of economic integration. Finally, the rich country never acts as both a tax and a pollution haven at equilibrium while the poor country can act as both. Indeed, when both capital mobility and domestic environmental awareness are low, the poor country can benefit from cutting corporate taxes to increase its attractiveness. The paper is organized as follows. The next section outlines the related literature.

Section 3 presents the main characteristics of the model. Section 4 defines the subgame perfect Nash equilibria of the sequential game. Section 5 provides some comparative statics. Section 6 deals with extensions of our model. Section 7 discusses the model and its policy-making implications.

2 Related literature

To the best of our knowledge, our paper is the first to jointly tackle tax and environmental competition between a rich and a poor country in a very simple framework where firms are imperfectly mobile and countries differ in their levels of environmental concern. However, there is a long-standing strand of the theoretical public finance literature in which fiscal competition is modeled as a non-cooperative game where public authorities tax mobile capital and provide public goods. One of the main results is that equilibrium tax rates are too low to adequately finance public goods. The standard model of tax competition has been extended in various directions (see Keen and Konrad (2013) for a comprehensive survey), including through the combination of tax and environmental competition between identical countries. This small section of the literature dates back to the seminal paper by Oates and Schwab (1988), which showed that governments set inefficiently low capital taxes at equilibrium and relax their environmental standards. In an influential paper, Ogawa and Wildasin (2009) challenge this result and find that tax competition for mobile capital leads to a first-best outcome regardless of transboundary pollution and hence, that there is no need for international environmental policy coordination. However, the latter result has recently been put into question by Yamagishi (2019), who emphasizes that Ogawa and Wildasin (2009)'s efficiency result crucially depends on the assumption that the level of environmental standards is exogenous – or put differently, that governments have no say in environmental regulations. Relaxing this assumption leads back to the more intuitive result that competition between countries or regions leads to weak environmental standards at equilibrium. Neither paper considers differently sized or differently developed countries. Nor do they consider the effects of economic integration on equilibrium levels of corporate taxes and environmental standards, since standard tax competition models

assume that capital is perfectly mobile and boils down to a continuum of investors of insignificant size. Finally, countries are assumed to have the same environmental preferences, which of course is quite a stretch.²

Most of the tax competition literature deals with symmetric agents. Bucovetsky (1991) and Wilson (1991) are exceptions. They show that small countries, in terms of population, set lower capital tax rates than larger countries do. However, these models do not deal with environmental issues and assume that per capita endowments do not vary between countries. We, in contrast, assume that the countries are different: the density of firms is higher in the rich country than in the poor one. Other papers have looked beyond simple population asymmetry. For instance, Cai and Treisman (2005) model fiscal competition over productivity-enhancing infrastructure between a richly endowed country and a poorly endowed country. They show that international capital mobility leads to a situation in which the poor country drops out of the competition and the rich country becomes unconstrained. Capital mobility exacerbates initial inequalities and hinders economic development. However, their model assumes perfect capital mobility and does not consider environmental policies. Models that include new economic geography factors such as increasing returns to scale and transportation costs in a tax competition framework generally show that core (high market potential) countries set higher taxes than peripheral ones do, but that trade integration tends to reduce the tax gap (Baldwin and Krugman, 2004; Haufler and Wooton, 2010; Exbrayat and Geys, 2014). Once again, to the best of our knowledge, this literature ignores the interactions between tax and environmental competition. The framework of our model is closer to the literature in which fiscal competition is modeled as a two-stage game where governments compete in the first stage over a productivity-enhancing public infrastructure and, in the second stage, over a corporate tax (Justman et al., 2005; Bénassy-Quéré et al., 2007; Pieretti and Zanaj, 2011). This literature shows that the productivityenhancing infrastructure mitigates the intensity of tax competition. Besides, Pieretti and Zanaj (2011) interestingly find that for intermediate mobility costs, the smaller

²Environmental preferences are shaped by a variety of factors such as social norms, the economic and social development of a region, and green advocacy. See Glaeser (2014) for a deep analysis of the supply of environmentalism and to Bowles (1998) for the endogenous formation of environmental preferences. See also Elhadj and Tarola (2015) for a recent analysis of the relationship between economic and environmental preferences.

jurisdiction can attract foreign capital by providing a higher level of public goods than the larger one, rather than undercutting it.

The environmental economics literature has likewise mostly ignored the interactions between tax and environmental competition, focusing mainly on the optimal environmental policy in a strategic setting to address the issues of carbon leakage and pollution havens (Naegele and Zaklan, 2019). Some of these papers endogenize firms' choice of location as a reaction to exogenous unilateral pollution reduction measures (Markusen et al., 1993; Motta and Thisse, 1994; Petrakis and Xepapadeas, 2003; Sanna-Randaccio and Sestini, 2012; Sanna-Randaccio et al., 2017), while a different strand of the literature endogenizes both the environmental policy and the choice of location. However, most of these models assume that countries are perfectly symmetric, that pollution is local and that there are no transport costs (Ulph and Valentini, 2001; Abe and Zhao, 2005; Ikefuji et al., 2016). The key argument in this literature is that when jurisdictions compete for FDI, this leads to a race to the bottom in terms of environmental standards. The ongoing process of globalization encourages governments to attract multinational firms from elsewhere by lowering environmental standards (Kayalica and Lahiri, 2005). Polluting firms thereby move from countries with stricter regulations to countries with weaker environmental standards, leading to the formation of pollution havens. Of course, this disregards the effect of market size asymmetry, which may be a factor in attracting firms (Böhringer et al., 2014; Sanna-Randaccio et al., 2017).

From an empirical perspective, several studies have documented the detrimental effect of high corporate taxes on FDI (De Mooij et al., 2006; Feld and Heckemeyer, 2011). Furthermore, empirical estimates of tax reaction functions have found that countries mimic each other's corporate taxation policies (Devereux et al., 2008; Overesch and Rincke, 2011; Redoano, 2014). As far as environmental regulations are concerned, some empirical studies have found a negative and potentially causal effect of the stringency and enforcement of environmental regulations on FDI decisions. However, this effect is small and often concentrated in the most pollution-intensive industries. Recently, Kellenberg (2009), Shahbaz et al. (2015), Sarkodie and Strezov (2019) and Yuan et al. (2019) confirmed the validity of the pollution haven hypothesis. However, other studies still question the fact that more stringent environmental standards really do lead to an outflow of firms (Kirkpatrick and Shimamoto, 2008; Kheder and Zugravu, 2012; Dou and Han, 2019).

The empirical literature has also revealed strategic interactions and mimicking behavior in the adoption and enforcement of environmental regulations between US States (Fredriksson et al., 2004) and between countries (Davies and Naughton, 2014). Note finally that in our model, countries compete over a bundle of instruments. Fredriksson et al. (2004) identify strategic interactions between US states on tax policy, productive public expenditures, and environmental regulations. More specifically, their results show strong evidence of interstate strategic interactions both within the scope of a given instrument and between different instruments. In other words, states may react to lax environmental standards in neighboring states by cutting taxes.

3 The model

We consider an economy composed of two countries, 1 and 2, with different densities of firms, $s_i \forall i = 1, 2$, and $s_1 + s_2 = 1$, so that $s_1 \geq \frac{1}{2}$. The countries can be nations, regions or sub-national jurisdictions. In the rest of the paper, we call the country with the higher initial density of firms (country 1) the "rich" country and country 2 the "poor" country. Our model can also be interpreted in terms of differences in population if we assume that the number of capital owners who are also entrepreneurs and workers is proportional to population size. As size can also be taken to mean geographical size, we describe the countries as "poor" and "rich" as this more clearly conveys the notion of wealth. The countries have tax autonomy and levy a corporate tax t_i . They also set environmental standards through the emissions cap α_i .

3.1 Firms

Each firm is run by a worker-entrepreneur and is endowed with one unit of capital. The fixed quantity q produced by each firm is sold on a competitive world market at a given price. Without loss of generality, we normalize the price to one.

Production of the output q is polluting. When producing in i, a firm incurs a cost, $C(q, \alpha_i, \bar{\alpha})$, with

$$C(q, \alpha_i, \bar{\alpha}) = c(q) + (F(\bar{\alpha}) + \mu(\bar{\alpha} - \alpha_i))$$

where α_i is the emissions cap in *i*, while $\bar{\alpha}$ represents the internationally-set emissions cap. The international limit is not controlled by country *i* but comes for example from an international agreement or from a higher level of government.³

The above function includes a variable cost of production, c(q), with $\frac{\partial C(q,\alpha_i,\bar{\alpha})}{\partial q} = \frac{\partial c(q)}{\partial q} > 0$, and a fixed cost of compliance with the emissions cap, $(F(\bar{\alpha}) + \mu(\bar{\alpha} - \alpha_i))$, which satisfies $\frac{\partial C(q,\alpha_i,\bar{\alpha})}{\partial \alpha_i} < 0$. The cost of compliance consists of two components: the first, $F(\bar{\alpha})$, with $\frac{\partial F(\bar{\alpha})}{\partial \bar{\alpha}} > 0$, captures firms' efforts to comply with the international cap, given their current technological capabilities. This could be investment in end-of-pipe measures or the implementation of cleaner production techniques.⁴ When the international and local emission caps coincide, the second part of the compliance cost $\mu(\bar{\alpha} - \alpha_i)$ disappears. However, when $\bar{\alpha} \neq \alpha_i$, the cost of compliance for firms located in country *i* is no longer fully captured by $F(\bar{\alpha})$. In particular, when the emissions cap in country *i* are laxer than the international norm $((\bar{\alpha} - \alpha_i) < 0)$, the cost for firms located in country *i* of complying with the local cap, α_i , is reduced by $\mu|\bar{\alpha} - \alpha_i|$. In contrast, in the case of more stringent local standards, the cost of compliance is increased by the same amount $\mu |\bar{\alpha} - \alpha_i|$. Without any loss of generality, we normalize μ to 1.

3.2 Firm location decisions

Firms are mobile and distributed over the interval [0, 1] in decreasing order of their willingness to invest abroad. The willingness to invest abroad of firm l, initially located in country i, is denoted by $x_{i,l}$. Following Pieretti and Zanaj (2011), we assume that relocating abroad costs firms a unit cost, k < 1.

³While the Kyoto protocol defined legally binding emission caps for developed countries, in the Paris Agreement, countries are no longer constrained by a globally shared cap. In particular, caps can be set at the regional or national level, so emissions are reduced on a voluntary basis.

⁴Incineration plants for waste disposal are a typical example of end-of-pipe technologies. In contrast, cleaner technologies reduce the environmental impact of production by fully or partially replacing polluting technologies. The use of environmentally friendly materials is a typical example of cleaner production measures (Frondel et al., 2007; Mantovani et al., 2017).

Thereby, if firm l remains in country i, its profits are given by:

$$\pi_{i,l}^i = q - C(q, \alpha_i, \bar{\alpha}) - t_i \quad \forall i = 1, 2$$

where t_i is the profit tax in country *i*.

Conversely, if firm l relocates from country i to country j, its profits are given by:

$$\pi_{i,l}^{j} = q - C(q, \alpha_{i}, \bar{\alpha}) - t_{j} - kx_{i,l} \quad \forall i = 1, 2 \quad j = 1, 2 \text{ and } i \neq j.$$

We assume that q is high enough to guarantee that if firm l relocates from i to j, its profits are non-negative, i.e. $q \ge C(q, \alpha_j) + t_j + kx_{i,l}$.

Defining x_1 as a neutral attitude toward relocating abroad from country 1, namely that firms with $x_{1,l} < x_1$ are willing to relocate from 1 to 2 while firms with a $x_{1,l} > x_1$ prefer not to, this threshold is defined as follows:

$$q - \bar{\alpha} + \alpha_1 - c(q) - F(\bar{\alpha}) - t_1 = q - \bar{\alpha} + \alpha_2 - c(q) - F(\bar{\alpha}) - t_2 - kx_1.$$
(1)

Symmetrically, considering a firm initially located in country 2 and indifferent to staying in 2 or relocating to 1 leads to $x_2 = -x_1$. To simplify the notation for the rest of the paper, we denote by $x = x_1 = -x_2$ the cut-off between staying put and relocating abroad. From (1) we obtain

$$x = \frac{1}{k} \left((t_1 - t_2) - (\alpha_1 - \alpha_2) \right)$$
(2)

When $(t_1 - t_2) > (\alpha_1 - \alpha_2)$, namely x > 0, there is a positive flow of firms from 1 to 2, while on the contrary, $(t_1 - t_2) > (\alpha_1 - \alpha_2)$ (x < 0) entails a flow of firms from 2 to 1. It immediately follows from (2) that

$$\frac{\partial x}{\partial t_1} = -\frac{\partial x}{\partial t_2} = \frac{1}{k}$$
 and $\frac{\partial x}{\partial \alpha_1} = -\frac{\partial x}{\partial \alpha_2} = -\frac{1}{k}$.

Any change in either the profit tax or the emissions cap in country i, has an effect on the number of firms relocating to country j, i.e. on x, which, unsurprisingly, is inversely related to the mobility cost k.⁵

3.3 Governments

The countries are assumed to maximize their tax revenues net of pollution-related disutility. Note that the payoff function allows the government to be both benevolent toward the environmental preferences of entrepreneur-workers and selfish through tax revenues.⁶

Formally, the payoff functions G_1 and G_2 of the public authorities in the two countries are written:

$$G_1(\alpha_1, t_1, \alpha_2, t_2) = R_1 - e_1 = s_1(1 - x)t_1 - e_1,$$

$$G_2(\alpha_1, t_1, \alpha_2, t_2) = R_2 - e_2 = ((1 - s_1) + s_1x)t_2 - e_2$$

with⁷

$$e_1 = \frac{1}{2} \left(\gamma_1 s_1 (1-x) \alpha_1^2 + \phi_1 (1-s_1+xs_1) \alpha_2^2 \right)$$

and

$$e_2 = \frac{1}{2} \left(\gamma_2 (1 - s_1 + s_1 x)) \alpha_2^2 + \phi_2 s_1 (1 - x) \alpha_1^2 \right).$$

where $s_1(1-x)$ and $1-s_1+s_1x$ are respectively the numbers of firms in country 1 (N_1) and country 2 (N_2) .

In the payoff function G_i , e_i represents the disutility of environmental damage in country

⁵Note that x, which represents a neutral attitude to staying put and relocating, also corresponds to the proportion of firms willing to relocate abroad.

⁶This payoff corresponds to a purely benevolent government objective in the case where tax revenues are redistributed through transfers to citizens. Our paper falls within the literature on governments that are neither purely benevolent nor purely Leviathan. See Edwards and Keen (1996) and Cai and Treisman (2005)

⁷Introducing the disutility of environmental damage as a convex function of the pollution level in the payoff function is in line with the standard literature on environmental economics. See for instance Baksi (2014), Bárcena-Ruiz (2006), Ulph (1996) and Falk and Mendelsohn (1993) for quadratic functions.

i. This damage consists of a *domestic component* (the first term in the brackets) and a *foreign component* (the second term in the brackets).⁸

The domestic component is proportional to the emissions generated in country *i*. In particular, it is determined (i) by the number of firms located in *i* at equilibrium and (ii) by the emissions cap in country *i*: the higher the value of α_i is, the laxer the environmental regulations are in this country and thus the higher the emissions generated by firms per unit of production are. Finally, the parameter γ_i captures the level of environmental concern in country *i* for local pollution, which depends on the specific set of environmental norms and values in that country.⁹

The foreign component, i.e. transboundary pollution, by symmetry with the above, is determined by (i) the number of firms located in the foreign country and (ii) the emissions cap in the foreign country, α_j . All other factors being equal, the impact that this component has on the environmental disutility, e_i , depends on ϕ_i , which reflects both the environmental awareness of country *i* and the type of transboundary pollution. Indeed, the impact of the transboundary pollution depends on the form it takes, for example acid rain, marine pollution or air quality degradation.

We do not make any assumptions about the relative strengths of the concerns about local and foreign pollution $(\gamma_i \gtrless \phi_i)$. At first sight, it may seem that an environmentally aware country should suffer more from domestic emissions than from cross-border pollution since the former have a direct impact while the latter only affects the country indirectly. Still, it could be that a country is bothered by a polluting neighbor to such an extent that transboundary pollution becomes a greater concern than domestic emissions (Unteroberdoerster, 2001).

⁸For country 1, the domestic component is $\gamma_1 \left(s_1(1-x)\alpha_1^2 \right)$ while the foreign component is $\phi_1 \left((1-s_1+xs_1)\alpha_2^2 \right)$.

⁹The policy-maker's concerns about pollution may be driven by pure or impure altruism. The environment can be considered a public good to be protected (Andreoni, 1990). Alternatively, emissions reduction can be seen as a reputational driver (Benabou and Tirole, 2006) or be motivated by moral concerns (Frey, 1999).

The disutility of environmental damage depends on the flow of firms as follows:

$$\frac{\partial e_1}{\partial x} = \frac{1}{2} s_1 (\phi_1 \alpha_2^2 - \gamma_1 \alpha_1^2) \stackrel{\geq}{\equiv} 0 \Longleftrightarrow \frac{\phi_1}{\gamma_1} \stackrel{\geq}{\equiv} \left(\frac{\alpha_1}{\alpha_2}\right)^2 \tag{3}$$

$$\frac{\partial e_2}{\partial x} = \frac{1}{2} s_1 (\gamma_2 \alpha_2^2 - \phi_2 \alpha_1^2) \stackrel{\geq}{\equiv} 0 \Longleftrightarrow \frac{\phi_2}{\gamma_2} \stackrel{\leq}{\equiv} \left(\frac{\alpha_2}{\alpha_1}\right)^2 \tag{4}$$

Accordingly, a reduction in the number of firms in country 1 (i.e. an increase in x) increases the environmental disutility for any $\frac{\phi_1}{\gamma_1} > \left(\frac{\alpha_1}{\alpha_2}\right)^2$ *i.e.* when the concern for transboundary pollution relative to local pollution is higher than a threshold that depends on the ratio of the emissions caps in the two countries. A decrease in the number of firms in country 1 leads, all other factors being equal, to lower local emissions and higher transboundary pollution is far higher than it is for local pollution, it may well be that the environmental disutility suffered by country 1 actually increases when the number of firms it hosts decreases. This is undoubtedly the case when the environmental standards in 2 are lower than in 1: a laxer environmental policy in the foreign country combined with a higher concern for transboundary over local pollution means that the environmental disutility increases when firms leave the country. This corresponds to a "yes in my backyard" (YIMBY) mechanism, the opposite of the well-known NIMBY ("not it my backyard") phenomenon, whereby the presence of firms in the country is considered environmentally favorable.

Conversely, when $\phi_1 < \gamma_1$, if the environmental regulations in country 1 are weaker than in country 2, i.e. $\alpha_1 > \alpha_2$, the environmental disutility in country 1 always decreases as x increases. Indeed, if country 1 sets a higher emissions cap than country 2 and is highly intolerant of local pollution, an outflow of firms is environmentally beneficial from country 1's perspective. The relocation of firms from country 1 to country 2 leads to a decrease in the disutility due to domestic pollution that is not completely offset by the effects of transboundary pollution from country 2. This corresponds to a NIMBY mechanism, since an outflow of firms from the country is considered environmentally beneficial.¹⁰ This generates opposition to the presence of firms in this country, since

¹⁰NIMBY projects are all undertakings that increase the overall payoff (tax revenues) but impose net costs on individuals in the host community (a private bad, such as pollution).

they are considered undesirable.

By following the same argument, any decrease in x, meaning an increase in the number of firms in country 1, increases the environmental disutility if the concern for local pollution is substantially higher than it is for transboundary pollution.

The above analysis only concerns the effects of x on environmental disutility. For the sake of completeness, it is worth noting that although firms do generate domestic pollution, they also determine the tax base of a country, and the larger the tax-base, the higher the corporate tax revenue (all other things being equal). The *net impact* of hosting firms in a country accounts for these two competing effects.

4 Sequential game

The two governments play a two-stage game. First, they choose an emissions cap α_i and second, they set their corporate tax rate. This sequence of decisions reflects the fact that environmental standards are long term decisions whereas taxes can vary from on year to another.

We assume that the mobility costs are not prohibitive, so that $x \neq 0$ for asymmetric countries. Solving the sequential game backward, we first solve the tax competition subgame.

Definition 1. The subgame perfect Nash equilibrium is defined as $(t_1^*(k, s_1), t_2^*(k, s_1), \alpha_1^*(k, s_1), \alpha_2^*(k, s_1))$.

4.1 Tax competition subgame

The second stage is solved under the assumption that equilibrium emissions caps have been defined in the first stage by the two countries. Maximizing $G_1(\alpha_1, t_1, \alpha_2, t_2)$ and $G_2(\alpha_1, t_1, \alpha_2, t_2)$ w.r.t t_1 and t_2 yields

$$\frac{dG_i}{dt_i} = \frac{dR_i}{dt_i} - \frac{de_i}{dt_i} = 0$$

with

$$\frac{dR_1}{dt_1} = \underbrace{s_1}^{\text{tax level effect}} - \underbrace{s_1\left(x + \frac{t_1}{k}\right)}_{s_1\left(x + \frac{t_1}{k}\right)} = \frac{1}{k}s_1\left(k + \alpha_1 - \alpha_2 - 2t_1 + t_2\right)$$

and
$$\frac{\partial e_1}{\partial t_1} = \frac{1}{k}\frac{\partial e_1}{\partial x} = \frac{1}{k}\frac{1}{2}s_1(\phi_1\alpha_2^2 - \gamma_1\alpha_1^2)$$

and for country 2

$$\frac{dR_2}{dt_2} = \underbrace{1-s_1}^{\text{tax level effect}} + \underbrace{s_1\left(x - \frac{t_2}{k}\right)}_{k} = 1 - \frac{1}{k}s_1\left(k + \alpha_1 - \alpha_2 - t_1 + 2t_2\right)$$

and
$$\frac{\partial e_2}{\partial t_2} = -\frac{1}{k}\frac{\partial e_2}{\partial x} = -\frac{1}{k}\frac{1}{2}s_1(\gamma_2\alpha_2^2 - \phi_2\alpha_1^2)$$

which respectively give the following best response functions:

$$t_1(t_2) = \frac{1}{4} \left(2(k+t_2) + 2(\alpha_1 - \alpha_2) + \gamma_1 \alpha_1^2 - \phi_1 \alpha_2^2 \right)$$

$$t_2(t_1) = \frac{(2k(1-s_1) + 2s_1t_1 + s_1 (2(\alpha_2 - \alpha_1) + \gamma_2 \alpha_2^2 - \phi_2 \alpha_1^2))}{4s_1}$$

The slopes of the two tax response functions are positive and lower than 1. In line with the tax competition literature, the corporate tax rates are strategic complements.¹¹ There is thus a single set of subgame equilibrium tax rates, i.e.

$$t_1^*(\alpha_1, \alpha_2) = \frac{2k + 2ks_1 + 2s_1(\alpha_1 - \alpha_2) + \alpha_1^2 s_1(2\gamma_1 - \phi_2) + \alpha_2^2 s_1(\gamma_2 - 2\phi_1)}{6s_1},$$

$$t_2^*(\alpha_1, \alpha_2) = \frac{4k - 2ks_1 + 2s_1(\alpha_2 - \alpha_1) + \alpha_1^2 s_1(\gamma_1 - 2\phi_2) + \alpha_2^2 s_1(2\gamma_2 - \phi_1)}{6s_1}.$$

Moreover, we observe that

¹¹Most papers in both the theoretical and the empirical literature show that tax rates are strategic complements. For a recent survey on whether taxes are strategic complements or strategic substitutes, see for instance Vrijburg and de Mooij (2016).

$$\frac{\partial t_i^*(\alpha_i, \alpha_j)}{\partial \alpha_i} = \frac{1}{3} \left(\alpha_i (2\gamma_i - \phi_j) + 1 \right) \stackrel{\geq}{=} 0 \Leftrightarrow \gamma_i \stackrel{\geq}{=} \frac{1}{2} \left(\phi_j - \frac{1}{\alpha_i} \right)$$
(5)

$$\frac{\partial t_i^*(\alpha_i, \alpha_j)}{\partial \alpha_j} = \frac{1}{3} \left(\alpha_j (\gamma_j - 2\phi_i) - 1 \right) \stackrel{\geq}{=} 0 \Leftrightarrow \phi_i \stackrel{\leq}{=} \frac{1}{2} \left(\gamma_j - \frac{1}{\alpha_j} \right)$$
(6)

Expressions (5) and (6) account for the effects of the emissions caps on the tax competition subgame.

and

The effect of α_i on $t_i^*(\alpha_i, \alpha_j)$, measured by Equation(5), is driven by two factors: an environmental factor and a tax-revenue factor. First, an increase in α_i , for a given number of firms in *i*, directly increases domestic pollution. A higher α_i also makes country *i* more attractive for firms, lowering *x* and further increasing local emissions. To limit this increase, country *i* has the incentive to increase its corporate tax rate to limit the inflow of firms, which is environmentally beneficial for a sufficiently high value of γ_i , i.e. $\gamma_i > \frac{1}{2} \left(\phi_j - \frac{1}{\alpha_i} \right)$. Still, this limits the tax base, with a negative effect on tax revenue, all other things held equal. Compensating for this loss of revenue is a further incentive for country *i* to increase its corporate tax rate. As a result, $t_i^*(\alpha_i, \alpha_j)$ increases with α_i . Note that there is no incentive to limit the inflow of firms from country *j* if γ_i is low, i.e. $\gamma_i \leq \frac{1}{2} \left(\phi_j - \frac{1}{\alpha_i} \right)$. In these circumstances, since country *i* is less intolerant of local pollution, it has no reason to increase its corporate tax rate to limit the inflow of firms: the equilibrium tax t_i^* decreases as α_i increases, with a positive effect on the tax base.

Equation (6) shows that $t_i^*(\alpha_i, \alpha_j)$ and α_j can be substitutes or complements depending on the value of ϕ_i . This relationship stems from the fact that the presence of firms in a country has two environmental effects: one through local emissions and the other through transboundary pollution. If the concern for transboundary emissions (ϕ_i) is high, country *i* has an incentive to attract firms from *j* to both mitigate transboundary pollution and increase its corporate tax base. However, this increases domestic pollution in country *i*. The higher ϕ_i is, the lower the opportunity cost is for country *i* to attract firms in terms of environmental disutility (yes in my backyard effect). In this case, t_i and α_j are strategic substitutes. In contrast, if ϕ_i is low, t_i and α_j turn out to be complements. Indeed, when ϕ_i is low, transboundary pollution from country j is of little consequence. Accordingly, if country j relaxes its environmental standards, country i has no incentive to cut its corporate tax rate to attract firms (*not in my backyard* effect). Attracting firms in this way would increase the tax base but at the expense of increasing local pollution and possibly, of reducing tax revenue if the increase in the tax base does not compensate for the lower tax rate.

It is worth noting that

$$\frac{\partial t_i^*(\alpha_i, \alpha_j)}{\partial \gamma_i} = \frac{1}{3}\alpha_i^2 > 0 \quad \text{and} \quad \frac{\partial t_i^*(\alpha_i, \alpha_j)}{\partial \gamma_j} = \frac{1}{6}\alpha_j^2 > 0 \tag{7}$$

$$\frac{\partial t_i^*(\alpha_i, \alpha_j)}{\partial \phi_i} = -\frac{1}{3}\alpha_j^2 < 0 \quad \text{and} \quad \frac{\partial t_i^*(\alpha_i, \alpha_j)}{\partial \phi_j} = -\frac{1}{6}\alpha_i^2 < 0.$$
(8)

The equilibrium tax rate $t_i^*(\alpha_i, \alpha_j)$ increases with the level of concern about domestic pollution: a greater concern for domestic pollution reduces the incentive to attract firms, limiting government *i*'s ability to increase its corporate tax rate without suffering from an outflow of firms. The loss of tax revenue is compensated by the decrease in pollution. Since t_i and t_j are strategic complements, t_i also increases with the level of concern for domestic pollution in country *j*.

Conversely, increases in ϕ_i and ϕ_j both decrease the equilibrium rates in the tax game. Let us first consider the effect of ϕ_i on $t_i^*(\alpha_i, \alpha_j)$: all other things being equal, an increase in ϕ_i increases the negative impact in country *i* of transboundary pollution from country *j*. If country *i* cuts its own tax rate to attract firms from country *j*, thereby limiting transboundary pollution, the inflow of firms will increase local emissions. Still, this negative environmental effect is weaker than the positive effect of increasing the tax base. Overall, this combination of factors leads to a "yes in my backyard effect".

The effect of ϕ_j on $t_i^*(\alpha_i, \alpha_j)$ is indirect. It is mediated by $t_j^*(\alpha_i, \alpha_j)$, which decreases as ϕ_j increases as described above. Since $t_i^*(\alpha_i, \alpha_j)$ and $t_j^*(\alpha_i, \alpha_j)$ are strategic complements, $t_i^*(\alpha_i, \alpha_j)$ also decreases.

4.2 Emissions caps

We can now move to the first stage of the game. Plugging $t_1^*(\alpha_1, \alpha_2)$ and $t_2^*(\alpha_1, \alpha_2)$ into the payoff functions G_1 and G_2 , and maximizing $G_i(\alpha_i, \alpha_j)$ w.r.t. α_i , we obtain

$$\alpha_1^* = \frac{1}{\phi_2 + \gamma_1}$$
 and $\alpha_2^* = \frac{1}{\phi_1 + \gamma_2}$,

with

$$\alpha_1^* \stackrel{\geq}{\equiv} \alpha_2^* \iff (\gamma_1 + \phi_2) \stackrel{\leq}{\equiv} (\gamma_2 + \phi_1) \tag{9}$$

Interestingly, at equilibrium, the emissions cap α_i^* chosen by country *i* depends on its level of concern for domestic pollution (γ_i) and country j's level of concern for transboundary pollution (ϕ_i) . In contrast, the concern in country i about foreign pollution does not alter its equilibrium choice of emissions cap α_i^* . Contrary to the situation in a purely non-cooperative game, the sequential game involving taxes and emissions caps leads the two countries to internalize the effects of both their environmental policy and their rival's in the tax competition stage. While the effect of concerns about domestic pollution is quite intuitive (higher γ_i means lower α_i), the effect of the competing country's concerns about foreign pollution is less obvious (higher ϕ_j means lower α_i). When choosing α_i , country *i* takes into account the fact that its own emissions cap will trigger a reaction from country j. This reaction comes in the form of tax competition. From (8), a higher concern ϕ_j for transboundary emissions leads country *i* to reduce its corporate tax rate $t_i^*(\alpha_i, \alpha_j)$. This attracts firms, thereby reducing transboundary pollution, and enlarges country i's tax base, with a positive effect on the tax revenue. On the other hand, it increases the level of local emissions. To reduce the disutility due to local pollution, country i decreases its environmental standards α_i^* . The magnitude of the strategic effect of α_i^* on $t_i^*(\alpha_i, \alpha_j)$ is captured by (5): the greater the concern about transboundary pollution (ϕ_i) is, the weaker the interplay between environmental and tax policies $(\alpha_i \text{ and } t_i)$ is.

We can immediately derive the following result:

Proposition 2. When $\Delta_i > \Delta_j$, country j is a pollution haven $(\alpha_j^* > \alpha_i^*)$, with $\Delta_i = \gamma_i - \phi_i \ \forall i \ and \ i \neq j$. Let us recall that when setting their environmental standards, the two countries take into account the corresponding domestic and foreign emissions.¹² This occurs through the only transmission channel between the two countries which is the flow of firms, x. A net outflow of firms from a country reduces domestic emissions but increases cross-border pollution. What matters then is the difference between the two pollution concern parameters, let us call this the *concern gap for domestic/transboundary pollution*: $\Delta_i = \gamma_i - \phi_i \,\forall i$. When $\Delta_i > \Delta_j$, the pollution concern gap is more damaging in country i than in country j. This pushes country i to set a lower emissions cap than country j's, discouraging firms from producing in country i but limiting domestic pollution. Since country j has a smaller pollution concern gap, it is more inclined to set a higher emissions cap, which attracts firms and enlarges the tax base, but also increases domestic pollution.

4.3 Subgame perfect Nash equilibrium

The subgame perfect Nash equilibrium, defined as $(t_1^*(k, s_1), t_2^*(k, s_1), \alpha_1^*(k, s_1), \alpha_2^*(k, s_1), x_1^*(k, s_1))$ is given by,

$$t_1^* = \frac{s_1 + 1}{3s_1}k + \Phi_1 \text{ and } t_2^* = \frac{2 - s_1}{3s_1}k + \Phi_2$$

 $\alpha_1^* = \frac{1}{\phi_2 + \gamma_1} \text{ and } \alpha_2^* = \frac{1}{\phi_1 + \gamma_2}$

where $\Phi_i \equiv \frac{1}{6} \left(\frac{4\gamma_i + \phi_j}{(\gamma_i + \phi_j)^2} - \frac{4\phi_i + \gamma_j}{(\gamma_j + \phi_i)^2} \right) i, j = 1, 2 \text{ and } i \neq j.$

This configuration yields the equilibrium flow of firms:

$$x^*(s_1, \gamma_1, \phi_1) = \frac{1}{6} \left(\frac{2(2s_1 - 1)}{s_1} + \frac{(\gamma_1 - \phi_1) - (\gamma_2 - \phi_2)}{k(\phi_1 + \gamma_2)(\phi_2 + \gamma_1)} \right)$$
(10)

with $x^* < 1$ iff $k > k_{Min} \equiv \frac{s_1((\gamma_1 - \phi_1) - (\gamma_2 - \phi_2))}{2(s_1 + 1)(\phi_2 + \gamma_1)(\phi_1 + \gamma_2)}$, whereas $x^* > -1$ if and only if $k > k_{Max} \equiv \frac{1}{2} s_1 \frac{(\gamma_2 - \phi_2) - (\gamma_1 - \phi_1)}{(\phi_2 + 1)(\phi_1 + 1)(5s_1 - 1)}$.¹³

 $^{^{12}}$ For the sake of simplicity, let us call the country that strategically sets a higher emissions cap than its rival a pollution haven, and the country that sets the lower corporate tax rate a tax haven.

¹³Note that $k_{Max} < 0$ for $\phi_2 > \phi_1$ and $k_{Min} < 0$ for $\phi_1 > \phi_2$.

Note that the equilibrium flow of firms x^* depends on the two sources of asymmetry we consider in our model. The first source, captured by the term $\frac{2(2s_1-1)}{s_1}$ is the asymmetry in firm densities between the two countries. The higher density of firms in country 1 $(s_1 \ge \frac{1}{2})$ implies that at equilibrium, firms flow from the rich country to the poor one. Because of its initially larger tax base, the former benefits more from increasing its tax rate than the latter does. The countries are also asymmetric in their levels of concern about pollution, captured by γ_i and ϕ_i , i = 1, 2. When the pollution concern gap is higher in the rich country than in the poor one, $(\gamma_1 - \phi_1) > (\gamma_2 - \phi_2)$, then the equilibrium standards chosen by country 1 are lower than country 2's, increasing the a net flow of firms towards country 2.

At equilibrium, the number of firms in country 1 is:

$$N_1^* = s_1(1 - x^*) = \frac{s_1((\gamma_2 - \phi_2) - (\gamma_1 - \phi_1))}{6ks_1(\phi_1 + \gamma_2)(\phi_2 + \gamma_1)} + \frac{s_1 + 1}{3s_1}$$
(11)

and in country 2:

$$N_2^* = 1 - s_1 + s_1 x^* = \frac{s_1((\gamma_1 - \phi_1) - (\gamma_2 - \phi_2))}{6ks_1(\phi_1 + \gamma_2)(\phi_2 + \gamma_1)} + \frac{2 - s_1}{3s_1}$$
(12)

Finally, the difference in payoffs at equilibrium is

$$G_1^* - G_2^* = \frac{k(2s_1 - 1)}{3s_1} + \frac{2\gamma_1 + 5\phi_2}{6(\gamma_1 + \phi_2)^2} - \frac{2\gamma_2 + 5\phi_1}{6(\gamma_2 + \phi_1)^2}$$
(13)

Equations (10) to (13) characterize the perfect subgame equilibrium as a function of the mobility costs k and the pollution concern gaps $\Delta_i \forall i$. Table 1 in the appendix sums up the different cases, and we can derive the following propositions.

Proposition 3. The poor country is unambiguously a tax haven if $\Delta_2 > \Delta_1$. If $\Delta_2 < \Delta_1$, the poor country is a tax haven if and only if capital mobility is sufficiently low $(k \ge \tilde{k})$.

In line with the standard literature on tax havens, the poor country is inclined to lower its corporate tax rate to attract firms and increase its tax revenue. Indeed, the rich country suffers less from an outflow of firms than the poor one does. This argument is in line with Kanbur and Keen (1993) and Pieretti and Zanaj (2011), among others. When $\Delta_2 > \Delta_1$, since the pollution concern gap is larger in the poor country, it sets stricter environmental regulations (proposition 2) thereby weakening the incentive for firms to leave country 1. To mitigate this effect, the poor country decreases its corporate tax rate as a means of attracting firms. This result does not hold when the poor country's pollution concern gap is smaller than its rival's, i.e. when $\Delta_2 < \Delta_1$. In this case, the poor country sets a higher emissions cap than its rival's (see proposition 2), increasing the outflow of firms from the latter. Which of the two countries sets the lower tax rate depends on capital mobility. When capital mobility is low, i.e. $k \geq \tilde{k}$, firms are less responsive to weak environmental policies and the poor country has to increase its attractiveness by ensuring its corporate tax rate is lower. On the contrary, when capital mobility is high, i.e. $k < \tilde{k}$, firms are more responsive to an asymmetry in environmental policies and the outflow of firms from the outflow of firms from country 1 is high enough that the poor country does not need a generous tax policy to be attractive. The poor country can set a higher corporate tax rate than the rich country's $(t_1^* < t_2^*)$.

The following corollary of Propositions 2 and 3 completes this specific case:

Corollary 4. The rich country only chooses to be a tax haven if the poor country acts as a pollution haven.

This corresponds to the case in which international mobility costs are low $(k < \tilde{k})$. Indeed, when countries are highly integrated (low capital mobility costs), firms are highly reactive to any gap in environmental standards and the rich country has to cut its tax rate to limit the relocation of firms to the poor country.

In the more restrictive case of very low mobility costs $(k < \hat{k} < \tilde{k})$, the high outflow of firms from 1 to 2 leads to there being more firms in 2 than in 1 at equilibrium: $N_2^* > N_1^*$. This finding holds in spite of the lower initial density of firms in the poor country.

Combining Propositions 2 and 3, we also obtain the conditions under which the poor country acts as both a tax and a pollution haven:

Corollary 5. The poor country is both a tax haven and a pollution haven when $\Delta_2 < \Delta_1$

and international capital mobility is low $(k > \tilde{k})$.

If the poor country is less concerned about generating pollution than the rich country is $(\Delta_2 < \Delta_1)$, it relaxes its environmental standards to attract firms (Proposition 2). Moreover, if capital mobility is low, this relaxes tax competition and corporate tax rates increase in both countries. However, the tax rate is higher in country 1 because it can afford a small outflow of firms from its larger initial tax base: the negative effect of this outflow is outbalanced by the higher tax revenues levied on the remaining firms. In contrast, the poor country's interest is always to undercut the rich country's corporate tax rate, even when mobility costs are high and few firms relocate, to generate tax revenue. The high mobility costs restrict the flow of firms so that even though the poor country is a capital importer, the stock of firms remains higher in country 1: $N_1^* > N_2^*$.

Corollary 6. The rich country is never both a pollution haven and a tax haven.

This follows directly from Propositions 3 and 4. This corollary emphasizes the fact that the poor country is the only one with the incentive to be both a tax and a pollution haven. Indeed, because of its high initial density of firms, the rich country is less affected by an outflow of firms than the poor country is. When the rich country is a pollution haven, its environmental standards are attractive to firms and the country does not need to reinforce its attractiveness by setting a lower corporate tax rate. The holding force of the environmental policy is high enough to enable country 1 to set a higher corporate tax rate than country 2's.

The previous propositions and corollaries have depicted the strategic positions of the two countries (tax or pollution haven) and the consequences in terms of the flow and density of firms at equilibrium. We now turn to the the payoffs $(G_1^* \text{ and } G_2^*)$.

Proposition 7. Whatever the equilibrium configuration, the poor country's payoff can only ever be higher than the rich country's if the two economies are highly integrated $(k < \check{k})$.

As shown in the appendix, the expression for \check{k} is rather complex and cannot be ranked intuitively with respect to the other thresholds.¹⁴ Notice that \check{k} can be negative,

¹⁴The relative value of \check{k} with respect to the other thresholds $(\tilde{k}, \hat{k}, \text{ and } \bar{k})$ depends on the values of $(\gamma_1 + \phi_2)$ relative to $(\gamma_2 + \phi_1)$, of ϕ_1 relative to ϕ_2 , and of γ_1 relative to γ_2 .

in which case it is impossible for the poor country to do better than the rich one.

In particular, Proposition 7 does not exclude the possibility, when capital mobility is low, that even when the poor country acts simultaneously as a tax haven and a pollution haven, it still has a lower payoff than the rich country. The efforts undertaken to attract firms are so strong in terms of tax competition and pollution that they prevent the poor country from doing better than the rich one. A higher payoff for the poor country is even more unlikely if the poor country is less concerned about pollution (net effect).¹⁵ This result contradicts the commonly held idea that being a tax haven is beneficial for countries. Conversely, when mobility costs are low, being either a tax haven or a pollution haven allows the poor country to obtain the higher payoff at equilibrium. Hence, capital mobility is the key element that determines which country does better; a rich country that is a capital exporter and not a tax or a pollution haven can have the higher payoff when capital mobility is low.

5 Comparative statics

As a natural continuation of the above analysis, we now investigate how the configuration of the market changes with the parameters of the model.

First of all, it is easy to see that the initial density of firms in the rich country, s_1 , has an unambiguously negative effect on the corporate tax rates,

$$\frac{dt_1^*}{ds_1} = -\frac{1}{3}\frac{k}{s_1^2} < 0 \qquad \frac{dt_2^*}{ds_1} = -\frac{2}{3}\frac{k}{s_1^2} < 0$$

and no effect on the emissions caps at equilibrium. When the density of firms in country 1 increases (with a symmetric reduction in the density of firms in country 2), this directly decreases the tax base in country 2. Country 2 is induced to cut its tax rate to attract firms. Through a standard tax competition mechanism (taxes are strategic complements), country 1 reacts by cutting its own tax rate. However, because of the lower density of firms in country 2, the decrease is more substantial in 2 than it is in 1. This explains the positive effect of s_1 on x^* , which describes either an increase in the

¹⁵This happens when mobility costs are neither too low nor too high : $\tilde{k} < k < \check{k}$.

outflow of firms from 1 ($x^* > 0$) or a decrease in the inflow of firms to 1 ($x^* < 0$):

$$\frac{\partial x^*}{\partial s_1} = \frac{1}{3s_1^2} > 0.$$

Moreover, we observe that:

Proposition 8. As international capital mobility increases,

i) the tax gap between the rich and the poor country becomes narrower if $t_1^* > t_2^*$;

ii) the environmental standards set at equilibrium are not affected;

iii) the difference in payoffs between the rich and the poor country decreases when $G_1^* > G_2^*$.

Proof. Differentiating the equilibrium values with respect to k yields: $\frac{\partial t_1^*}{\partial k} = \frac{s_1+1}{3s_1} > 0$ and $\frac{\partial t_2^*}{\partial k} = \frac{2-s_1}{3s_1} > 0$. Moreover, $\frac{\partial t_1^*}{\partial k} - \frac{\partial t_2^*}{\partial k} = \frac{1}{3}\frac{2s_1-1}{s_1} > 0$ and $\frac{\partial (G_1^*-G_2^*)}{\partial k} = \frac{2s_1-1}{3s_1} > 0$ for any $s_1 \ge \frac{1}{2}$.

Note that this result is analogous to the effect of trade costs in trade models: when trade costs increase (decrease), the tax gap increases (decreases).

Both t_i^* and t_2^* increase with k but to a different extent. Because of the larger initial density of firms in country 1, the flow of firms is more sensitive to the level of capital integration (k) in country 1 than in country 2, and the corporate tax t_1^* is more responsive to changes in mobility costs. This stronger reaction means that the payoff of the rich country is more strongly affected by a change in k than the poor country's payoff is. More precisely, an increase in capital mobility reduces the difference in payoffs if the rich country has the higher payoff, but increases the difference if the rich country has the lower payoff at equilibrium (i.e. when mobility costs are already low: $k < \check{k}$). Finally, the level of capital integration does not affect the optimal emissions caps since capital mobility costs are fully integrated into the tax choices in this sequential game.

Note moreover that an increase in capital integration leads the rich country to become a larger capital exporter (when $x^* > 0$) or the poor country to become a smaller capital exporter (when $x^* < 0$) when $\Delta_2 > \Delta_1$, i.e. when the pollution concern gap is larger in the poor country than in the rich one:

$$\frac{\partial x^*}{\partial k} = \frac{1}{6} \frac{(\gamma_2 - \phi_2) - (\gamma_1 - \phi_1)}{k^2 (\phi_2 + \gamma_1) (\phi_1 + \gamma_2)} \stackrel{\geq}{\gtrless} 0 \Leftrightarrow \Delta_2 \stackrel{\geq}{\gtrless} \Delta_1.$$

As already mentioned, the concern for pollution leads to NIMBYism or YIMBYism depending on the concern gap for domestic/foreign pollution. When both countries have the the same pollution concern gap, the equilibrium flow of firms no longer depends on capital integration (k) since the flow's response to a change in capital integration is already taken into account in setting the corporate tax rate.

Finally, analyzing the effect of the levels of concern about pollution yields:

$$\frac{\partial t_i^*}{\partial \phi_i} = \frac{(2\phi_i - \gamma_j)}{3(\phi_i + \gamma_j)^3} \gtrless 0 \Rightarrow \phi_i \gtrless \frac{\gamma_j}{2} \quad \text{and} \quad \frac{\partial t_i^*}{\partial \phi_j} = -\frac{1}{6} \frac{\phi_j + 7\gamma_i}{(\phi_j + \gamma_i)^3} < 0$$
$$\frac{\partial t_i^*}{\partial \gamma_j} = \frac{1}{6} \frac{\gamma_j + 7\phi_i}{(\phi_j + \gamma_i)^3} > 0 \quad \text{and} \quad \frac{\partial t_i^*}{\partial \gamma_i} = \frac{(\phi_j - 2\gamma_i)}{3(\phi_j + \gamma_i)^3} \gtrless 0 \Rightarrow \gamma_i \nleq \frac{\phi_j}{2}$$

whereas

$$\frac{dx^{*}}{d\phi_{2}} = \frac{dx^{*}}{d\gamma_{1}} = \frac{1}{6k(\phi_{2} + \gamma_{1})^{2}} > 0 \quad \text{and} \quad \frac{dx^{*}}{d\phi_{1}} = \frac{dx^{*}}{d\gamma_{2}} = -\frac{1}{6k(\phi_{1} + \gamma_{2})^{2}} < 0$$

To grasp the economic rationale for this, recall that whenever γ_i and/or ϕ_i increase, all other factors being equal, this increases the disutility of local and/or transboundary pollution in country 1.

If the poor country's level of concern about the rich country's pollution is less than half what it is about its own pollution ($\phi_2 < \frac{\gamma_1}{2}$), country 2 reduces its tax rate t_2^* in response to a rise in ϕ_2 to attract firms. Indeed, an increase in ϕ_2 for a given γ_2 reduces the relative disutility of domestic pollution in country 2. Moreover, the benefit for country 2 of hosting firms, in terms of tax revenue, outbalances the disutility of the environmental damage these firms create. Country 1 reacts in two ways. On the one hand, it reduces its tax rate to limit the outflow of firms, and on the other, to reduced the aggressiveness of country 2's tax policy, it reduces its emissions cap. The emissions cap α_1^* is a means of reducing the disutility of transboundary pollution.

If ϕ_2 is high (i.e. $\phi_2 > \gamma_1/2$), any further increase substantially increases the disutility of transboundary pollution in country 2, which increases its tax rate (t_2^*) because the benefit of increasing its share of firms in terms of tax-revenue does not outbalance the environmental disutility it would suffer otherwise. In line with the above, country 1 reacts in two ways: it reduces its corporate tax rate to limit the outflow of firms and makes its environmental regulations more stringent to soften the reaction of country 2. The role of γ_i follows from the same argument.

6 Particular cases and discussion

We have derived our results in a general framework allowing for mobile firms and several kinds of asymmetries. In this section, we investigate the specific cases of symmetric countries and immobile firms. Finally, we extend our framework by considering asymmetric productivities.

6.1 Case of symmetric countries

The results for symmetric countries follow naturally from those with asymmetric countries. The countries can be partially or perfectly symmetric, i.e. symmetric in their firm densities, environmental awareness, or both.

When their levels of environmental concern are symmetric, i.e. $\gamma_i = \gamma_j = \gamma$ and $\phi_i = \phi_j = \phi$, it holds that $\alpha_1^* = \alpha_2^* = \frac{1}{\gamma + \phi}$ and $t_1^* \ge t_2^{*,16}$ Thus, the two countries set the same emissions cap at equilibrium and the poor country always acts as a tax haven. Moreover, at equilibrium, the flow of firms depends only on the countries' tax policies. The poor country is a capital importer but does worse than the rich country $(G_1^* > G_2^* \text{ because } \check{k} = 0)$. Thus, for identical levels of environmental concern (both countries suffer the same disutility from one unit of emissions, both for domestic and transboundary pollution), the poor country acts as a tax haven to attract firms and benefits from an increase in tax revenue that outbalances the negative effects of increased domestic pollution.

¹⁶The threshold mobility cost for tax equality is zero: $\tilde{k} = 0$ (see appendix).

The following proposition describes the outcome when the countries have the same initial density of firms:

Proposition 9. If the two countries have the same density of firms $(s_1 = s_2 = 1/2)$ but different pollution concern gaps $(\Delta_i \neq \Delta_j)$, at equilibrium, with $\Delta_i < \Delta_j$, country *i* is a pollution haven and country *j* a tax haven.

Notice that this result does not depend on the mobility costs k even though x^* depends on k. At equilibrium, the value of the policy tool depends on the relative levels of environmental concern. The country with the small pollution concern gap (country i) acts as a pollution haven, but its rival (country j) acts as a tax haven. The difference in emissions caps ($\alpha_i^* > \alpha_j^*$) drives the firms toward the country with the laxer environmental standards. The other country sets its tax rate lower than its rival's to limit the outflow of firms.

Obviously, when $\phi_i = \phi_j$ and $s_1 = s_2 = 1/2$, it follows that $\alpha_i^* = \alpha_j^* = \frac{1}{\gamma + \phi}$, $t_i^* = t_j^* = k + \frac{\gamma - \phi}{2(\gamma + \phi)^2}$ and $x^* = 0$. Stronger concerns about pollution generate more stringent environmental standards but lower corporate taxes because of more intense tax competition to attract firms.

6.2 Case of immobile firms

The case of immobile firms can be treated by assuming that the mobility costs k tend to $+\infty$. Thus, the condition expressed by (1) boils down to x = 0. With no capital mobility, the payoff function G_i is monotonically increasing in $t_i(\alpha_i, \alpha_j)$ since there is no tax competition mechanism with which to attract firms. This implies that taxes are set to the maximum level, i.e. the level at which profits are zero:¹⁷

 $t_i^*(\alpha_i) = q - c(q) - (K(\bar{\alpha}) + (\bar{\alpha} - \alpha_i)) \quad \forall i = 1, 2$

¹⁷Remember that we have set $\mu = 1$.

In the second stage, the optimal emissions cap is determined by two opposite drivers. On the one hand, the higher α_i is, the higher profits are, and the higher the corporate tax rate t_i and tax revenue are. On the other hand, a higher α_i also increases the disutility of environmental damage since in spite of the very stringent regulations in country i, this does not lead to an outflow of firms. At equilibrium, the optimal standards are:

$$\alpha_i^* = \frac{1}{\gamma_i} \quad \forall i = 1, 2.$$

The equilibrium is thus given by

$$(\alpha_1^*, t_1^*, \alpha_2^*, t_2^*) = \left(\frac{1}{\gamma_1}, q - c(q) - \left(K(\bar{\alpha}) + \left(\bar{\alpha} - \frac{1}{\gamma_1}\right)\right), \frac{1}{\gamma_2}, q - c(q) - \left(K(\bar{\alpha}) + \left(\bar{\alpha} - \frac{1}{\gamma_2}\right)\right)\right)$$

The equilibrium value α_i^* is easily derived because the profit function is linear with respect to the environmental standards. Nonetheless, the above equilibrium configuration provides interesting insights when compared with the one that emerges when firms are mobile. When firms are immobile, the emissions caps are not affected either by the tax rate in the competing country or by transboundary pollution, since environmental regulations no longer influence the allocation of firms between the countries. The only trade-off for public authorities is the arbitrage between the benefits of tax revenue (the corporate tax rate and environmental standards are complements) and the disutility of domestic pollution. The result is that in the absence of an internalization mechanism for the effects of transboundary pollution through the flow of firms, environmental regulations are weaker and the corporate tax rate is maximal for the chosen level of environmental standards.

6.3 Asymmetric productivities

For the sake of simplicity, the model considered so far assumes that the countries differ in wealth (density of firms) but not in productivity. We now extend our model by including the fact that country 1's output is greater than country 2's $(q_1 > q_2)$. This market power does not affect the environmental regulations in the two countries because the latter only depend on the levels of concern for pollution (because of the sequence of the game). Contrary to the environmental policies, the corporate tax rates are affected by the production gap so that $t_1^{*'} = t_1^* + \frac{Q}{3}$ and $t_2^{*'} = t_2^* - \frac{Q}{3}$, denoting the new equilibrium values with a prime and setting $Q = q_1 - c(q_1) - (q_2 - c(q_2))$ (calculations available on request from the authors). When Q > 0, which is the case for $c'(q_i) < 1$, the higher level of production in country 1 enables the government to set a higher tax rate without increasing the outflow of firms because at equilibrium, the capital outflow from 1 to 2 (or inflow from 2 to 1) is lower (higher) than when the countries have equal productivities: $x' = x - \frac{Q}{3k}$. Unsurprisingly, the equilibrium number of firms in country 1 is larger than with symmetric productivities $(N_1^{*'} = N_1^* + \frac{s_1}{3k})$ while it is lower in country 2 $(N_2^{*'} = N_2^* - \frac{s_1}{3k})$, which increases the rich country's payoff at the expense of the poor country's $(G_1^{*'} - G_2^{*'} = G_1^* - G_2^* + \frac{2Q}{3})$.

7 Concluding remarks and policy-making implications

Our main finding is the following: when setting their environmental standards, countries internalize the effect of their own pollution on the corporate tax rate set by the competing country. As a result, the equilibrium environmental standards depend on the levels of concern for both domestic and cross-border pollution. Furthermore, while equilibrium environmental standards are not affected by capital integration, corporate tax rates are .

The mechanisms governing our model may help explain several stylized facts about the interactions between interjurisdictional tax competition and environmental awareness. The following discussion must be considered suggestive. First, our model shows that poor or small countries have incentives to undercut rich and large countries and to some extent act as tax havens. This is in line with both the standard tax competition literature and the literature on tax competition and new economic geography. There is also evidence that on average, industrialized (OECD) countries have higher corporate income tax rates than non-OECD countries, although the difference has been narrowing over the past decade (Crivelli et al., 2015; Asen, 2019). On the same note, in the European Union (EU), the average statutory corporate tax rate is lower in Central and Eastern European Countries (CEEC) than in the EU15 as a whole, with mean tax rates

of respectively 17% in the CEEC and 24% in the EU15 in 2019.

Second, our paper explicitly models interactions between corporate taxation and environmental standards in a competitive setting, which seems a plausible assumption as there is empirical evidence that governments compete on several fronts and that they may react strategically via one instrument, say by cutting taxes, when faced with lax regulations in a competing state. More interestingly, our model shows that cutting corporate taxes and relaxing environmental standards is not the only possible strategy to attract firms, especially for large and industrialized countries. Developed countries can do better even if they have higher corporate taxes and high environmental standards. Because they are less sensitive to tax-induced offshoring than less developed countries, they can enjoy both high tax revenues and high environmental quality. Interestingly, it seems that economic integration may help bring developing countries closer to this situation as it narrows the tax gap between rich and poor countries. More generally, our model illustrates and explains the wide range of strategies available to governments (Figure 1 displays these strategies for EU-27 member states and the UK).¹⁸ Transition and developing countries are not all pollution havens and tax havens even if they are more likely to be pollution havens than industrialized countries are (Koźluk and Timiliotis, 2016).

Third, our model also suggests that even if stringent environmental standards may negatively affect firms' profits and dissuade them from basing operations in countries with a high environmental awareness, policy-makers choose pollution standards independently from international mobility costs and, more generally, from the degree of economic integration. This finding contrasts with the traditional view that globalization is likely to lead to a race-to-the bottom for corporate taxes and to a general relaxation of environmental standards. In our model, policy-makers set environmental standards based only on concerns about pollution. To some extent, policy-makers set environmental standards in keeping with their own environmental preferences while adjusting corporate tax rates to attract firms. This result is along the lines of Ogawa and Wildasin's 2009, even if our model differs from theirs in many respects. Our analysis suggests that coordination efforts should focus on tax rather than environmental policies, or at least, that

¹⁸The same kind of exercise can be carried for OECD and non-OECD countries.



Figure 1: Interplay between tax policies and environmental regulations in a sample of EU member states (Source: OECD and World Economic Forum)

leaving governments some leeway in setting their environmental regulations is possible. Our result is also in line with the large empirical literature that finds that corporate taxation is a strong factor in FDI decisions while the host country's environmental norms have a much smaller effect.

8 Appendix

8.1 Environmental standards

Plugging $t_1^*(\alpha_1, \alpha_2)$ and $t_2^*(\alpha_1, \alpha_2)$ into the payoff functions G_1 and G_2 , we get

$$\frac{dG_i}{d\alpha_i} = \frac{\partial G_i}{\partial \alpha_i} + \overbrace{\frac{\partial G_i}{\partial t_i^*(\alpha_1, \alpha_2)}}^{=0} \frac{\partial t_i^*(\alpha_1, \alpha_2)}{\partial \alpha_i} + \frac{\partial G_i}{\partial t_j^*(\alpha_1, \alpha_2)} \frac{\partial t_j^*(\alpha_1, \alpha_2)}{\partial \alpha_i} = 0$$

which reduces to:

$$\frac{dG_i}{d\alpha_i} = \frac{\left(\alpha_i(\phi_j + \gamma_i) - 1\right)\left(s_i\left(2(\alpha_j - \alpha_i) + \alpha_i^2(\phi_j + \gamma_i) - \alpha_j^2(\phi_i + \gamma_j)\right) - 2k(s_i + 1)\right)}{9k} = 0$$

Checking for the second order condition (i.e maximum), we obtain:

$$\frac{d^2 G_i}{d\alpha_i^2} = \frac{1}{9k} \left(3s_i (\gamma_i + \phi_j)^2 \alpha_i^2 - 6s_i (\gamma_i + \phi_j) \alpha_i - 2(\phi_j + \gamma_i) k(s_i + 1) - s_i \alpha_j^2 (\phi_j + \gamma_i) (\phi_i + \gamma_j) + 2\alpha_j (\phi_j + \gamma_i) s_i + 2s_i \right) \right)$$

Let us define the polynomial $P_1(\alpha_i) = a\alpha_i^2 + b\alpha_i + c$ with

$$a = \frac{(\phi_j + \gamma_i)^2 s_i}{3k} > 0$$

$$b = -\frac{2(\phi_j + \gamma_i) s_i}{3k} < 0$$

$$c = \frac{-2(\phi_j + \gamma_i)k(s_i + 1) - s_i \alpha_j^2(\phi_j + \gamma_i)(\phi_i + \gamma_j) + 2\alpha_j(\phi_j + \gamma_i)s_i + 2s_i}{9k}$$

To ensure that the solution to the FOC is a maximum, we first have to check that the discriminant is positive. We can easily derive that the discriminant is positive for $c < \frac{s_i}{3k}$, otherwise, the polynomial $P_1(\alpha_i)$ is positive for any α_i (since $c > \frac{s_i}{3k} > 0$) and the solution is a minimum.

Let us consider the case in which the discriminant is positive:

$$c < \frac{s_i}{3k} \Longleftrightarrow k > \frac{2\alpha_j(\phi_j + \gamma_i)s_i - s_i\alpha_j^2(\phi_j + \gamma_i)(\phi_i + \gamma_j) - s_i}{2(\phi_j + \gamma_i)(s_i + 1)}$$

We obtain the two following roots for the polynomial :

$$\overline{\alpha}_{i} = \frac{s_{i} + \sqrt{\frac{s_{i}\left(2(\phi_{j}+\gamma_{i})k(s_{i}+1)+s_{i}\left(\alpha_{j}^{2}(\phi_{j}+\gamma_{i})(\phi_{i}+\gamma_{j})-2\alpha_{j}(\phi_{j}+\gamma_{i})+1\right)\right)}{3}}{(\phi_{j}+\gamma_{i})s_{i}}$$

$$s_{i} - \sqrt{\frac{s_{i}\left(2(\phi_{j}+\gamma_{i})k(s_{i}+1)+s_{i}\left(\alpha_{j}^{2}(\phi_{j}+\gamma_{i})(\phi_{i}+\gamma_{j})-2\alpha_{j}(\phi_{j}+\gamma_{i})+1\right)\right)}{3}}{3}}$$

 $\underline{\alpha}_{i} = \frac{s_{i} - \sqrt{3}}{(\phi_{j} + \gamma_{i})s_{i}}$ Since $P_{1}\left(\frac{1}{\gamma_{i} + \phi_{j}}\right) = c - \frac{s_{i}}{3k} < 0$ we have the following condition:

For
$$k > \frac{2\alpha_j(\phi_j + \gamma_i)s_i - s_i\alpha_j^2(\phi_j + \gamma_i)(\phi_i + \gamma_j) - s_i}{2(\phi_j + \gamma_i)(s_i + 1)}$$
, $P_1(\alpha_i) < 0 \iff \alpha_i \in [\underline{\alpha}_i, \overline{\alpha}_i]$.

From Equation 8.1, we obtain 3 solutions:

$$\begin{aligned} \alpha_{i1} &= \frac{1}{\phi_j + \gamma_i} \quad with \quad \overline{\alpha}_i > \alpha_{i1} > \underline{\alpha}_i \\ \alpha_{i2} &= \frac{s_i + \sqrt{s_i \left(2(\phi_j + \gamma_i)k(s_i + 1) + s_i \left(\alpha_j^2(\phi_j + \gamma_i)(\phi_i + \gamma_j) - 2\alpha_j(\phi_j + \gamma_i) + 1\right)\right)}}{(\phi_j + 1)s_i} > \overline{\alpha}_i \\ \alpha_{i3} &= \frac{s_i - \sqrt{s_i \left(2(\phi_j + \gamma_i)k(s_i + 1) + s_i \left(\alpha_j^2(\phi_j + \gamma_i)(\phi_i + \gamma_j) - 2\alpha_j(\phi_j + \gamma_i) + 1\right)\right)}}{(\phi_j + \gamma_i)s_i} < \underline{\alpha}_i \end{aligned}$$

The concavity condition excludes solutions 2 and 3, which are minima, and solution 1 is the only maximum: $\alpha_i = \frac{1}{\phi_j + \gamma_i}$ for $k > \frac{s_i((\gamma_i - \phi_i) - (\gamma_j - \phi_j))}{2(\gamma_j + \phi_i)(\gamma_i + \phi_j)(1 + s_i)}$.

8.2 Equilibrium configurations

The values of the threshold mobility costs are:

$$\begin{split} \tilde{k} &\equiv \frac{5((\gamma_1 - \phi_1) - (\gamma_2 - \phi_2))s_1}{2(2s_1 - 1)(\gamma_2 + \phi_1)(\gamma_1 + \phi_2)}, \text{ and } t_1^* - t_2^* = \frac{2k(2s_1 - 1)(\gamma_2 + \phi_1)(\gamma_1 + \phi_2) + 5s_1(\phi_1 - \phi_2)}{6s_1(1 + \phi_1)(1 + \phi_2)} > 0 \text{ iff } k > \tilde{k}. \\ \bar{k} &\equiv \frac{s_1((\gamma_2 - \phi_2) - (\gamma_1 - \phi_1))}{(\phi_2 + \gamma_1)(\phi_1 + \gamma_2)(2s_1 - 1)}, \text{ and } x^* > 0 \iff k > \bar{k} \text{ from (10)}, \\ \hat{k} &= \frac{s_1((\gamma_1 - \phi_1) - (\gamma_2 - \phi_2)}{(2s_1 - 1)(\gamma_2 + \phi_1)(\gamma_1 + \phi_2)} \text{ and } N_1^* > N_2^* \iff k > \hat{k}. \\ \text{Finally, } \check{k} &\equiv \frac{3s_1}{(2s_1 - 1)} \left(\frac{2\gamma_2 + 5\phi_1}{6(\gamma_2 + \phi_1)^2} - \frac{2\gamma_1 + 5\phi_2}{6(\gamma_1 + \phi_2)^2} \right) \text{ and } G_1^* - G_2^* > 0 \iff k > \check{k}. \end{split}$$

Note that:

	$oldsymbol{\Delta_2} > oldsymbol{\Delta_1}$		$\Delta_1 > \Delta_2$		
	$\bar{k} > 0$		$\tilde{k} > \hat{k} > 0$		
	$\tilde{k} < 0$ and $\hat{k} < 0$		$\bar{k} < 0$		
	$\alpha_1^* > \alpha_2^*$		$\alpha_1^* < \alpha_2^*$		
$\forall k$	$t_1^* > t_2^*$		$x^* > 0$		
	$N_{1}^{*} > N_{2}^{*}$				
$k > \check{k}$	$k > \bar{k}$	$k < \bar{k}$	$k > \tilde{k}$	$k < \tilde{k}$	
	$x^* > 0$	$x^* < 0$	$t_1^* > t_2^*$	$t_1^* < t_2^*$	
				for $\tilde{k} > k > \hat{k}$	for $\tilde{k} > \hat{k} > k$
			$N_1^* > N_2^*$	$N_1^* > N_2^*$	$N_1^* < N_2^*$
	$G_1^* > G_2^*$		$G_{1}^{*} > G_{2}^{*}$		
	$k > \bar{k}$	$k < \bar{k}$	$k > \tilde{k}$	$k < \tilde{k}$	
$k < \check{k}$	$x^* > 0$	$x^* < 0$	$t_1^* > t_2^*$	$t_1^* < t_2^*$	
				for $\tilde{k} > k > \hat{k}$	for $\tilde{k} > \hat{k} > k$
			$N_1^* > N_2^*$	$N_1^* > N_2^*$	$N_1^* < N_2^*$
	$\overline{G_1^* < G_2^*}$		$G_{1}^{*} < G_{2}^{*}$		

Table 1: Sets of equilibria depending on the level of capital integration and the ranking of pollution concern gaps for $k > k_{Max}$ and $k > k_{Min}$ with $\Delta_i = \gamma_i - \phi_i$.

$$\tilde{k} \equiv \frac{5((\gamma_1 - \phi_1) - (\gamma_2 - \phi_2))s_1}{2(2s_1 - 1)(\gamma_2 + \phi_1)(\gamma_1 + \phi_2)} > 0 \Leftrightarrow (\gamma_1 - \phi_1) > (\gamma_2 - \phi_2).$$

$$\bar{k} \equiv \frac{s_1((\gamma_2 - \phi_2) - (\gamma_1 - \phi_1))}{(\phi_2 + \gamma_1)(\phi_1 + \gamma_2)2(2s_1 - 1)} > 0 \Leftrightarrow (\gamma_2 - \phi_2) > (\gamma_1 - \phi_1)$$

$$\hat{k} = \frac{s_1((\gamma_1 - \phi_1) - (\gamma_2 - \phi_2))}{(2s_1 - 1)(\gamma_2 + \phi_1)(\gamma_1 + \phi_2)} > 0 \Leftrightarrow (\gamma_1 - \phi_1) > (\gamma_2 - \phi_2)$$

The sign of \check{k} is much more complicated to determine and depends on the ranking of $(\gamma_1 - \phi_1)$ and $(\gamma_2 - \phi_2)$ and on the ranking of ϕ_1 and ϕ_2

Let us rank the threshold values \hat{k}, \check{k} and \tilde{k} , when positive:

$$\begin{split} \tilde{k} - \check{k} &\equiv \frac{3}{2} s_1 \frac{\gamma_2 (\gamma_1 + \phi_2)^2 - \gamma_1 (\gamma_2 + \phi_1)^2}{(\phi_2 + 1)^2 (\phi_1 + 1)^2 (2s_1 - 1)} \stackrel{\geq}{\gtrless} 0 \Leftrightarrow \gamma_2 (\gamma_1 + \phi_2)^2 \stackrel{\leq}{\lessgtr} \gamma_1 (\gamma_2 + \phi_1)^2 \\ \hat{k} - \check{k} &\equiv \frac{3}{2} s_1 \frac{\phi_1 (\gamma_2 + \phi_1)^2 - \phi_2 (\gamma_1 + \phi_2)^2}{(\phi_2 + 1)^2 (\phi_1 + 1)^2 (2s_1 - 1)} \stackrel{\geq}{\gtrless} 0 \Leftrightarrow \phi_1 (\gamma_2 + \phi_1)^2 \stackrel{\geq}{\gtrless} \phi_2 (\gamma_1 + \phi_2)^2 \\ \tilde{k} - \hat{k} &= 3 s_1 \frac{(\gamma_1 - \phi_1) - (\gamma_2 - \phi_2)}{(2s_1 - 1) (\phi_2 + 1) (\phi_1 + 1)} \stackrel{\geq}{\gtrless} 0 \Leftrightarrow \gamma_1 + \phi_2 \stackrel{\leq}{\lessgtr} \gamma_2 + \phi_1 \end{split}$$

$$\bar{k} - \check{k} = \frac{3s_1}{2(2s_1 - 1)} \left(\frac{\gamma_1 + 2\phi_2}{(\phi_2 + \gamma_1)^2} - \frac{\gamma_2 + 2\phi_1}{(\phi_1 + \gamma_2)^2} \right) \stackrel{\geq}{\equiv} 0 \Leftrightarrow \frac{\gamma_1 + 2\phi_2}{(\phi_2 + \gamma_1)^2} \stackrel{\geq}{\equiv} \frac{\gamma_2 + 2\phi_1}{(\phi_1 + \gamma_2)^2}$$

So whenever

(i)
$$\gamma_2 - \phi_2 > \gamma_1 - \phi_1$$
, it follows $\bar{k} > 0$, $\hat{k} < 0$ and $\tilde{k} < 0$
(ii) $\gamma_1 - \phi_1 > \gamma_2 - \phi_2$, it holds that $\tilde{k} > \hat{k} > 0$ and $\bar{k} < 0$.

Moreover, we know that $k_{Max} < 0$ for $\gamma_1 - \phi_1 > \gamma_2 - \phi_2$ and $k_{Min} < 0$ for $\gamma_2 - \phi_2 > \gamma_1 - \phi_1$. Let us check the ranking of k_{Max} and \bar{k} when $\gamma_2 - \phi_2 > \gamma_1 - \phi_1$ and the ranking of k_{Min} and \hat{k} when $\gamma_1 - \phi_1 > \gamma_2 - \phi_2$:

• Let us compare \bar{k} and k_{Max} :

$$\bar{k} - k_{\max} = \frac{3s_1^2}{2(2s_1 - 1)(5s_1 - 1)} \frac{(\gamma_2 - \phi_2) - (\gamma_1 - \phi_1)}{(\gamma_1 + \phi_2)(\gamma_2 + \phi_1)}$$

Then $\bar{k} > k_{Max}$ when $\gamma_2 - \phi_2 > \gamma_1 - \phi_1$.

• Now let us compare \hat{k} and k_{Min} :

$$\hat{k} - k_{Min} = \frac{(\gamma_1 - \phi_1) - (\gamma_2 - \phi_2)}{(\gamma_2 + \phi_1)(\gamma_1 + \phi_2)} \left(\frac{3s_1}{2(s_1 + 1)(2s_1 - 1)}\right)$$

Then $\hat{k} > k_{Min}$ when $\gamma_1 - \phi_1 > \gamma_2 - \phi_2$.

References

- Abe, K. and Zhao, L. (2005). Endogenous international joint ventures and the environment. Journal of International Economics, 67(1):221–240.
- Andreoni, J. (1990). Impure altruism and donations to public goods: A theory of warm-glow giving. *The Economic Journal*, 100(401):464–477.
- Asen, E. (2019). Corporate tax rates around the world, 2019. Tax Foundation.
- Baksi, S. (2014). Regional versus multilateral trade liberalization, environmental taxation, and welfare. Canadian Journal of Economics/Revue Canadienne d'économique, 47(1):232–249.
- Baldwin, R. E. and Krugman, P. (2004). Agglomeration, integration and tax harmonisation. *European Economic Review*, 48(1):1–23.
- Bárcena-Ruiz, J. C. (2006). Environmental taxes and first-mover advantages. Environmental and Resource Economics, 35(1):19–39.
- Benabou, R. and Tirole, J. (2006). Belief in a just world and redistributive politics. The Quarterly Journal of Economics, 121(2):699–746.
- Bénassy-Quéré, A., Gobalraja, N., and Trannoy, A. (2007). Tax and public input competition. *Economic Policy*, 22(50):386–430.
- Böhringer, C., Fischer, C., and Rosendahl, K. E. (2014). Cost-effective unilateral climate policy design: Size matters. *Journal of Environmental Economics and Man*agement, 67(3):318–339.
- Bowles, S. (1998). Endogenous preferences: The cultural consequences of markets and other economic institutions. *Journal of Economic Literature*, 36(1):75–111.
- Bucovetsky, S. (1991). Asymmetric tax competition. *Journal of Urban Economics*, 30(2):167–181.
- Cai, H. and Treisman, D. (2005). Does competition for capital discipline governments? decentralization, globalization, and public policy. *American Economic Re*view, 95(3):817–830.

- Chen, X., Huang, B., and Lin, C.-T. (2019). Environmental awareness and environmental kuznets curve. *Economic Modelling*, 77:2–11.
- Crivelli, E., De Mooij, R. A., and Keen, M. M. (2015). Base erosion, profit shifting and developing countries. International Monetary Fund.
- Davies, R. B. and Naughton, H. T. (2014). Cooperation in environmental policy: A spatial approach. International Tax and Public Finance, 21(5):923–954.
- De Mooij, R. A., Ederveen, S., et al. (2006). What a difference does it make? understanding the empirical literature on taxation and international capital flows. Technical report, Directorate General Economic and Financial Affairs (DG ECFIN), European Commission.
- Devereux, M. P., Lockwood, B., and Redoano, M. (2008). Do countries compete over corporate tax rates? *Journal of Public Economics*, 92(5-6):1210–1235.
- Dou, J. and Han, X. (2019). How does the industry mobility affect pollution industry transfer in china: Empirical test on pollution haven hypothesis and porter hypothesis. *Journal of Cleaner Production*, 217:105–115.
- Edwards, J. and Keen, M. (1996). Tax competition and leviathan. European Economic Review, 40(1):113–134.
- Elhadj, N. B. and Tarola, O. (2015). Relative quality-related (dis) utility in a vertically differentiated oligopoly with an environmental externality. *Environment and Development Economics*, 20(3):354.
- Exbrayat, N. and Geys, B. (2014). Trade integration and corporate income tax differentials. *International Tax and Public Finance*, 21(2):298–323.
- Falk, I. and Mendelsohn, R. (1993). The economics of controlling stock pollutants: An efficient strategy for greenhouse gases. *Journal of Environmental Economics and Management*, 25(1):76–88.
- Feld, L. P. and Heckemeyer, J. H. (2011). Fdi and taxation: A meta-study. Journal of Economic Surveys, 25(2):233–272.

- Fredriksson, P. G., List, J. A., and Millimet, D. L. (2004). Chasing the smokestack: strategic policymaking with multiple instruments. *Regional Science and Urban Eco*nomics, 34(4):387–410.
- Frey, B. S. (1999). Morality and rationality in environmental policy. Journal of Consumer Policy, 22(4):395–417.
- Frondel, M., Horbach, J., and Rennings, K. (2007). End-of-pipe or cleaner production? an empirical comparison of environmental innovation decisions across oecd countries. *Business Strategy and the Environment*, 16(8):571–584.
- Glaeser, E. L. (2014). The supply of environmentalism: Psychological interventions and economics. *Review of Environmental Economics and Policy*, 8(2):208–229.
- Haufler, A. and Wooton, I. (2010). Competition for firms in an oligopolistic industry: The impact of economic integration. *Journal of International Economics*, 80(2):239– 248.
- Ikefuji, M., Itaya, J.-i., and Okamura, M. (2016). Optimal emission tax with endogenous location choice of duopolistic firms. *Environmental and Resource Economics*, 65(2):463–485.
- Justman, M., Thisse, J.-F., and Van Ypersele, T. (2005). Fiscal competition and regional differentiation. *Regional Science and Urban Economics*, 35(6):848–861.
- Kayalica, M. Ö. and Lahiri, S. (2005). Strategic environmental policies in the presence of foreign direct investment. *Environmental and Resource Economics*, 30(1):1–21.
- Keen, M. and Konrad, K. A. (2013). The theory of international tax competition and coordination. In *Handbook of Public Economics*, volume 5, pages 257–328. Elsevier.
- Kellenberg, D. K. (2009). An empirical investigation of the pollution haven effect with strategic environment and trade policy. *Journal of International Economics*, 78(2):242–255.
- Kheder, S. B. and Zugravu, N. (2012). Environmental regulation and french firms location abroad: An economic geography model in an international comparative study. *Ecological Economics*, 77:48–61.

- Kirkpatrick, C. and Shimamoto, K. (2008). The effect of environmental regulation on the locational choice of japanese foreign direct investment. Applied Economics, 40(11):1399–1409.
- Koźluk, T. and Timiliotis, C. (2016). Do environmental policies affect global value chains? OECD Working Paper.
- Mantovani, A., Tarola, O., and Vergari, C. (2017). End-of-pipe or cleaner production? how to go green in presence of income inequality and pro-environmental behavior. *Journal of Cleaner Production*, 160:71–82.
- Markusen, J. R., Morey, E. R., and Olewiler, N. D. (1993). Environmental policy when market structure and plant locations are endogenous. *Journal of Environmental Economics and Management*, 24(1):69–86.
- Millimet, D. L. (2013). Environmental federalism: a survey of the empirical literature. Case Western Reserve Law Review, 64:1669.
- Motta, M. and Thisse, J.-F. (1994). Does environmental dumping lead to delocation? European Economic Review, 38(3-4):563–576.
- Naegele, H. and Zaklan, A. (2019). Does the eu ets cause carbon leakage in european manufacturing? *Journal of Environmental Economics and Management*, 93:125–147.
- Oates, W. E. and Schwab, R. M. (1988). Economic competition among jurisdictions: efficiency enhancing or distortion inducing? *Journal of Public Economics*, 35(3):333– 354.
- Ogawa, H. and Wildasin, D. E. (2009). Think locally, act locally: Spillovers, spillbacks, and efficient decentralized policymaking. *American Economic Review*, 99(4):1206–17.
- Overesch, M. and Rincke, J. (2011). What drives corporate tax rates down? a reassessment of globalization, tax competition, and dynamic adjustment to shocks. *The Scandinavian Journal of Economics*, 113(3):579–602.
- Petrakis, E. and Xepapadeas, A. (2003). Location decisions of a polluting firm and the time consistency of environmental policy. *Resource and Energy Economics*, 25(2):197–214.

- Pieretti, P. and Zanaj, S. (2011). On tax competition, public goods provision and jurisdictions' size. *Journal of International Economics*, 84(1):124–130.
- Redoano, M. (2014). Tax competition among european countries. does the eu matter? European Journal of Political Economy, 34:353–371.
- Sanna-Randaccio, F. and Sestini, R. (2012). The impact of unilateral climate policy with endogenous plant location and market size asymmetry. *Review of International Economics*, 20(3):580–599.
- Sanna-Randaccio, F., Sestini, R., and Tarola, O. (2017). Unilateral climate policy and foreign direct investment with firm and country heterogeneity. *Environmental and Resource Economics*, 67(2):379–401.
- Sarkodie, S. A. and Strezov, V. (2019). A review on environmental kuznets curve hypothesis using bibliometric and meta-analysis. *Science of the Total Environment*, 649:128–145.
- Shahbaz, M., Nasreen, S., Abbas, F., and Anis, O. (2015). Does foreign direct investment impede environmental quality in high-, middle-, and low-income countries? *Energy Economics*, 51:275–287.
- Ulph, A. (1996). Environmental policy and international trade when governments and producers act strategically. *Journal of Environmental Economics and Management*, 30(3):265–281.
- Ulph, A. and Valentini, L. (2001). Is environmental dumping greater when plants are footloose? *Scandinavian Journal of Economics*, 103(4):673–688.
- Unteroberdoerster, O. (2001). Trade and transboundary pollution: Spatial separation reconsidered. Journal of Environmental Economics and Management, 41(3):269–285.
- Vrijburg, H. and de Mooij, R. A. (2016). Tax rates as strategic substitutes. International Tax and Public Finance, 23(1):2–24.
- Wilson, J. D. (1991). Tax competition with interregional differences in factor endowments. *Regional Science and Urban Economics*, 21(3):423–451.

- Yamagishi, A. (2019). Transboundary pollution, tax competition and the efficiency of uncoordinated environmental regulation. *Canadian Journal of Economics/Revue Canadienne d'économique*, 52(3):1165–1194.
- Yuan, F., Wei, Y. D., Gao, J., and Chen, W. (2019). Water crisis, environmental regulations and location dynamics of pollution-intensive industries in china: A study of the taihu lake watershed. *Journal of Cleaner Production*, 216:311–322.